Author's response to reviews

Title: Factors influencing the development of evidence-based practice among nurses: a self-report survey

Authors:

Anne Dalheim (anne.dalheim@helse-bergen.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Roy M Nilsen (roy.miodini.nilsen@mfr.uib.no)
Monica W Nortvedt (monica.nortvedt@hib.no)

Version: 5 Date: 12 October 2012

Author's response to reviews: see over
Author’s response to the editors.

Title: Factors influencing the development of evidence-based practice among nurses: a self-report survey

Authors:

Anne Dalheim (anne.dalheim@helse-bergen.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Roy Miodini Nilsen (roy.nilsen@mfr.uib.no)
Monica Wammen Nortvedt (monica.nortvedt@hib.no)

Version: 4 Date: October 12. 2012

Author’s response: see below
To the Editors of BMC Health Services Research

We appreciate the additional comments on our manuscript entitled “Factors influencing the development of evidence-based practice among nurses” (MS: 3079672846585478). Changes that have been made in response to the comments from the copyeditor have been written in “red” in the revised manuscript.

Sincerely yours,
Anne Dalheim, on behalf of the author group.